Author’s response to reviews

Title: Cross-validation of a Learning Climate Instrument in a non-Western Postgraduate Clinical Environment

Authors:

Jaime Pacifico (jaime_pacifico@yahoo.com; jlpacifico@dlshsi.edu.ph)

Cees van der Vleuten (c.vandervleuten@maastrichtuniversity.nl)

Arno Muijtjens (a.muijtjens@maastrichtuniversity.nl)

Erlyn Sana (erlynsana@yahoo.com.ph)

Sylvia Heeneman (s.heeneman@maastrichtuniversity.nl)

Version: 2 Date: 25 Sep 2017

Author’s response to reviews:

Corrections, revisions have been done based on the 4 technical comments.

These are in the Declarations part.